

Author's response to reviews

Title: Corruption, inequality and population perception of health system quality in Europe

Authors:

Zlatko Nikoloski (Z.Nikoloski@lse.ac.uk)
Elias Mossialos (E.Mossialos@lse.ac.uk)

Version: 3 **Date:** 7 January 2013

Author's response to reviews:

The authors are grateful for the editorial comments sent on 19th of December. As per the guidance for resubmission the following changes points of the article were addressed:

1. A title page corresponding to submission guidance was included in the beginning of the manuscript.
2. A competing interest section was also included in the manuscript.
3. Specific authors' contribution section was also added to the manuscript.
4. The details on the availability of data were also provided. Data is publicly available.

In addition to the changed made above, a section on abbreviations was also added to the manuscript.